

Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL089335

Key Points:

- Characterizing the storage potential of firn aquifers is important for predicting water budgets in a warming world
- Multiple geophysical methods are combined to evaluate the elastic properties and porosity of the firn aquifer
- Seismic velocities detect an increase in ice content in the aquifer downslope, reducing the porosity and storage potential of the firm

Supporting Information:

- Supporting Information S1
- Table S1
- Table S2
- Table S3
- Table S4

Correspondence to:

S. F. Killingbeck, eespr@leeds.ac.uk

Citation:

Killingbeck, S. F., Schmerr, N. C., Montgomery, L. N., Booth, A. D., Livermore, P. W., Guandique, J., et al. (2020). Integrated borehole, radar, and seismic velocity analysis reveals dynamic spatial variations within a firn aquifer in southeast Greenland. *Geophysical Research Letters*, 46, e2020GL089335. https://doi.org/ 10.1029/2020GL089335

Received 17 JUN 2020 Accepted 1 SEP 2020 Acceted article online 9 SEP 2020

$\textcircled{\sc c}2020.$ The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Integrated Borehole, Radar, and Seismic Velocity Analysis Reveals Dynamic Spatial Variations Within a Firn Aquifer in Southeast Greenland

S. F. Killingbeck¹ [b], N. C. Schmerr² [b], L. N. Montgomery³ [b], A. D. Booth¹ [b], P. W. Livermore¹ [b], J. Guandique², O. L. Miller⁴ [b], S. Burdick⁵, R. R. Forster⁶ [b], L. S. Koenig⁷ [b], A. Legchenko⁸ [b], S. R. M. Ligtenberg⁹ [b], C. Miège¹⁰ [b], D. K. Solomon¹¹ [b], and L. J. West¹

¹School of Earth and Environment, University of Leeds, Leeds, UK, ²Department of Geology, University of Maryland, College Park, MD, USA, ³Atmospheric and Oceanic Sciences Department, University of Colorado Boulder, Boulder, CO, USA, ⁴United States Geological Survey, Utah Water Science Center, West Valley City, UT, USA, ⁵Department of Geology, Wayne State University, Detroit, MI, USA, ⁶Department of Geography, University of Utah, Salt Lake City, UT, USA, ⁷National Snow and Ice Data Center, University of Colorado, Boulder, CO, USA, ⁸L'Institut des Géosciences de l'Environnement, IRD - Research Institute for Development, Marseille, France, ⁹Formerly Institute for Marine and Atmospheric Research Utrecht, Utrecht University, Utrecht, The Netherlands, ¹⁰Formerly Department of Geography, Rutgers, The State University of New Jersey, New Brunswick, NJ, USA, ¹¹Department of Geology and Geophysics, University of Utah, Salt Lake City, UT, USA

Abstract Perennial water storage in firn aquifers has been observed within the lower percolation zone of the southeast Greenland ice sheet. Spatially distributed seismic and radar observations, made ~50 km upstream of the Helheim Glacier terminus, reveal spatial variations of seismic velocity within a firn aquifer. From 1.65 to 1.8 km elevation, shear-wave velocity (*Vs*) is $1,290 \pm 180$ m/s in the unsaturated firn, decreasing below the water table (~15 m depth) to $1,130 \pm 250$ m/s. Below 1.65 km elevation, *Vs* in the saturated firn is $1,270 \pm 220$ m/s. The compressional-to-shear velocity ratio decreases in the downstream saturated zone, from 2.30 ± 0.54 to 2.01 ± 0.46 , closer to its value for pure ice (2.00). Consistent with colocated firn cores, these results imply an increasing concentration of ice in the downstream sites, reducing the porosity and storage potential of the firn likely caused by episodic melt and freeze during the evolution of the aquifer.

Plain Language Summary An integrated geophysical analysis of seismic, radar, and borehole measurements has been completed over a firn aquifer in southeast Greenland. We show the stiffness of the aquifer increases at lower elevations, closer to sea level, which leads to a decrease in pore space for the meltwater to be stored. This corresponds to an increase in ice content within the firn at lower elevations, as observed in borehole measurements, and likely caused by the meltwater refreezing within and below the aquifer.

1. Introduction

Warming of polar ice sheets changes the hydrological structure of surface firn and ice. Distribution and routing of surface meltwater to the base of an ice sheet can alter the sub- and en-glacial hydrologic systems and affect ice flow (e.g., Bartholomew et al., 2010). In particular, englacial water storage in firn aquifers, defined here as a storage of liquid water above the firn-ice transition (Fountain & Walder, 1998), can have contrasting implications on the influence of surface melting on ice sheet dynamics. Firn aquifers could either temporarily buffer the delivery of water to the oceans (Harper et al., 2012; Parry et al., 2007) or alternatively cause sudden release events when storage capacity is reached (Koenig et al., 2014). In Greenland, extensive firn aquifers have been observed in high-accumulation, high-melt (24–50 cm/yr; Miller et al., 2020) regions in the southeast in the last decade (e.g., Forster et al., 2014; Koenig et al., 2014; Kuipers Munneke et al., 2014; Miège et al., 2016). Meltwater stored within a firn aquifer is either drained from the system, though fracture/crevasse networks, or refreezes into ice. Refreezing densifies the firn column (Hubbard et al., 2016) and has been implicated as a precursor for the catastrophic collapse of Antarctic ice shelves (Kuipers Munneke et al., 2014). Densification also complicates the relationship between surface elevation change and changing ice mass, since an increase in surface density can cause a decrease in elevation (Reeh et al., 2005).



Figure 1. (a) Study location, ~50 km upstream of the Helheim Glacier terminus. Sites 1–12 (excluding 9) are seismic survey locations, with boreholes (FA16 4, 5, and 6) drilled in August 2016 at Sites 7, 8, and 12. The map hillshade and elevation contours are obtained from the ArcticDEM created by the Polar Geospatial Center from DigitalGlobe, Inc. imagery (Porter et al., 2018). (b) Elevation (above sea level) profile along the flow line and aquifer structure. (c) Seismic acquisition geometry used throughout. (d) Rayleigh wave dispersion curves for each site and (e) their implied depth sampling (Gazetas, 1982).

Therefore, knowledge of the geographic extent, storage capacity, water flow, and their temporal variations will enable an improved understanding of firn aquifers' impacts on glacial hydrology (Poinar et al., 2019), which, in turn, will be critical for accurately predicting the contribution of meltwater discharge and changing ice mass to global sea level rise.

Geophysical analysis is useful for quantifying firn hydrology without time-intensive drilling, providing the spatial distribution of firn properties with estimates of geophysical property uncertainty. Such quantitative

analysis can be improved, in terms of depth sensitivity and resolution, where independent geophysical observations are integrated into a joint analysis and combined with ground truth from cores or other point samples of aquifer thickness and hydraulic property measurements.

In 2011, a firn aquifer in the upstream region of Helheim Glacier (southeast Greenland) was studied with ice coring and ground-penetrating radar (GPR) methods (Forster et al., 2014; Koenig et al., 2014). Airborne radar data have been used to measure the depth and extent of the aquifer (e.g., Chu et al., 2018; Miège et al., 2016), but the utility of radar sampling beneath the aquifer is limited because liquid water increases radar attenuation rates. Magnetic resonance sounding (MRS) methods have been used to estimate the volume of water stored within the aquifer; however, vertical resolution and the aquifer base were poorly resolved (e.g., Legchenko et al., 2018a, 2018b). Point samples, provided by boreholes, directly measure the density structure of the aquifer and its detailed hydrological properties (Miller et al., 2018), but these do not provide the spatial coverage of geophysical surveying.

Seismic refraction profiling has been used to obtain the compressional (*P*-) wave velocity (*Vp*) structure of the firn and ice to improve the estimate of the aquifer-ice transition depth (e.g., Montgomery et al., 2017). However, thus far, there is no constraint of the shear (*S*-) wave velocity (*Vs*) distribution in a firn aquifer. *Vs* is related to the shear rigidity of a material, and the ratio Vp:*Vs* is a powerful indicator of consolidation and fluid content (Lee, 2002). Combined analysis of *Vp* and *Vs* could, therefore, improve discrimination between the saturated and unsaturated zones of the firn aquifer. The seismic data described in Montgomery et al. (2017) feature strong Rayleigh waves, a form of seismic energy often termed as "ground roll" that travels along the ground surface. While Rayleigh waves are typically considered noise in seismic refraction/reflection analysis, this wave component is useful to inform the *Vs* structure of the firn and ice (e.g., Diez et al., 2014; Young Kim et al., 2010).

In this paper, we present an integrated geophysical analysis of the firn aquifer on Helheim Glacier by deriving *Vs* from seismic Rayleigh waves, leveraging independent Vp, density, and thickness constraints using the "MuLTI" algorithm (Killingbeck et al., 2018) and estimating porosity of the aquifer using the derived Vp:Vsratios and Biot-Gassmann theory (Biot, 1941; Gassmann, 1951). We use the results to infer the spatial variability of seismic velocity and porosity within the aquifer. Our approach is a novel complement to the conventional geophysical analysis of firn aquifers and provides valuable constraints for modeling their formation.

2. Methods

2.1. Field Site and Data Acquisition

GPR, seismic, and firn cores were acquired along a flow line of Helheim Glacier, ~50 km upstream of its terminus, between July and August 2016 (Figure 1a) over a known firn aquifer (Forster et al., 2014; Miège et al., 2016). Active-source seismic data were acquired at 11 sites (numbered 1–12, excluding 9 where no seismic was acquired in 2016 and for consistency with Montgomery et al., 2017) along the flow line, and boreholes were drilled at Sites 7, 8, and 12. The water table depth was estimated as 10–20 m below the surface (Figure 1b) from colocated GPR data (Miège et al., 2016). The seismic acquisition is detailed in Montgomery et al., (2017), and their seismic refraction analyses suggest the aquifer base (firn-ice transition) to be ~28 m below the surface (Figure 1b). Borehole dilution tests (Miller et al., 2018) show that water flows through the aquifer with an average specific discharge of 4.3×10^{-6} m/s, decreasing to 0 m/s at the aquifer base. Miller et al. (2018) also show firn cores that record the density structure of the aquifer.

2.2. Rayleigh Wave Analysis

Multichannel analysis of surface waves (MASW; Park et al., 1999) is employed to estimate the Vs structure of the aquifer. Vs can be linked to the shear modulus (or stiffness), μ , and density, ρ , of a material by the following:

$$Vs = \sqrt{\frac{\mu}{\rho}}.$$
 (1)

Since the shear modulus of air and fluids is 0, the replacement of air by water in the saturated zone of the aquifer produces no change in stiffness; however, the same replacement increases the bulk density of the





Figure 2. (a) Zero-offset shot gather from Site 7. (b) Dispersion shown and picked in the frequency-phase velocity domain. (c) Posterior distributions of *Vs*, *Vp*, and ρ from MuLTI III (water table and aquifer base shown by the blue and blue-dashed lines). The mean of the prior and its standard deviation are shown by the red and red-dashed lines, respectively; the mode solutions are shown by the black lines. The gray shaded area highlights data outside the reliable depth range estimated (Gazetas (1982).

firn; hence, *Vs* would be expected to reduce if stiffness remained constant. Conversely, the presence of ice lenses in the aquifer increases its stiffness, therefore leading to an increase in *Vs*.

The zero-offset shot gathers, at both ends of Arrays 1 and 2 (Figure 1c), were transformed into the frequency-phase velocity domain, where the dispersive pattern of the Rayleigh waves was sampled (Park et al., 1999). The dispersion curves for each array are similar; hence, a homogenous layered subsurface was assumed for the underlying firn such that a single one-dimensional dispersion curve could be analyzed for each site. The dispersion curves suggest that shear-wave phase velocity increases downstream towards lower elevations (Figure 1d), while it decreases with depth through the aquifer (Figure 1e). The estimated depth range resolved by the Rayleigh wave dispersion curves is \sim 3–40 m, based on Gazetas' (1982) approximation that the depth sampled is one-third of the wavelength used.

2.3. Inverting for Vs

Rayleigh wave dispersion curves can be inverted to provide vertical *Vs* profiles. Our analysis is based on a probabilistic Bayesian tool, "MuLTI" (Killingbeck et al., 2018), that jointly inverts Rayleigh wave dispersion curves with layer-depth information into a depth-dependent posterior distribution of *Vs*. Here we extend the algorithm into a variant "MuLTI III" that additionally allows the inversion to be constrained using depth-dependent density and *Vp* profiles, thus integrating all available geophysical observations. The data used for our analysis include

- 1. The picked Rayleigh wave dispersion curves (Figure 1d) and their uncertainty, σ (Figure 2b),
- 2. The water table and aquifer base depths derived from GPR and seismic refraction, respectively (Figure 1b; Montgomery et al., 2017),
- 3. *Vp* profiles and their uncertainty σ_{Vp} derived from a Bayesian inversion of *P*-wave refractions (Montgomery et al., 2017),
- 4. Density profiles and their uncertainty σ_{ρ} from the firn cores (Miller et al., 2018).

Using a transdimensional Markov chain Monte Carlo method, we create a large ensemble drawn from the joint posterior distribution of the three parameters (*Vs*, *Vp*, ρ), with more likely models sampled with greater frequency. The posterior distribution is the product of the likelihood and the prior information. Here, the likelihood is based upon fitting theoretical dispersion curves to those observed, assuming a normal distribution with a mean given by the forward model calculated using the Geopsy algorithm (Wathelet et al., 2020) and standard deviation σ . The GPR and refraction data determine the vertical extent of three internal layers: unsaturated firn, aquifer, and ice. These data are also used to define the uniformly distributed prior distributions for *Vs* in each of the layers. The prior depth-dependent distributions for *Vp* and ρ are determined using seismic refraction (3) and density data (4), respectively, assuming normal distributions centered on their profile values with standard deviation σ_{Vp} and σ_{ρ} . The model is discretized using *k* piecewise constant functions based on Voronoi cells, where *k* is unknown but follows a distribution also sought by the method. Where acceptable in terms of likelihood, the method always chooses smooth (low *k*) models over rough models (high *k*).

Figure 2 shows an example of the surface wave analysis and output products from MuLTI III for Site 7, colocated with borehole measurements from FA16-4 (see supporting information). We obtain complete marginal probability density functions (PDFs) of *Vs*, *Vp*, and ρ for which we show the mode (highest probability) value and uncertainty as one standard deviation for each depth (Figure 2c). The posterior distributions of *Vp* and ρ are comparable to their prior distributions, showing that the data used in the joint inversion are consistent with datas ets (3) and (4) individually. MuLTI III was applied to all sites; where boreholes were absent, the average density of the 3 internal layers was calculated from borehole measurements and applied with an uncertainty of 10%, typical of the uncertainty in the firn cores and permitting variability within the model space.

3. Results

3.1. Profiles of Vs Distributions With Depth

The most likely *Vs* solutions are derived for all sites (Figure 3a). Average *Vs* values for the unsaturated firm above the water table (Layer 1), saturated aquifer (Layer 2), and underlying ice (Layer 3) are shown in Figure 3c. For Sites 1–7, located at higher elevations (>1.65 km) within the aquifer system, the average *Vs*





Figure 3. (a) Most-probable one-dimensional *Vs* distributions for all sites, output from MuLTI III. (b) Most-probable one-dimensional *Vp:Vs* distributions for all sites, output from MuLTI III. The full lines in these figures indicated the resolvable depth range of the Rayleigh wave inversion. (c) Average *Vs* structure and (d) average *Vp:Vs* structure for sites <1.65 km (1–7) and >1.65 km elevation (10–12). Layer 1 averages are calculated for unsaturated firm at depths >10 m; therefore, site 12 is not included in this average. All error bars represent one standard deviation.

in the unsaturated firn, at ~15–20 m depth, is 1,290 \pm 180 m/s. *Vs* decreases to 1,130 \pm 250 m/s through saturated firn at ~25 m depth and is as low as 990 \pm 270 m/s at Site 1, furthest upslope, in the saturated firn.

At lower elevations (<1.65 km), the aquifer maintains the same depth beneath the surface, but higher values of Vs are observed for the unsaturated $(1,420 \pm 200 \text{ m/s})$ and saturated $(1,265 \pm 220 \text{ m/s})$ firm (increases of 10% and 12%, respectively, vs. upstream sites). These changes are attributed to an increase in shear modulus (equation 1) caused by the introduction of clear ice lenses, increasing the ice content within the aquifer. Borehole observations at Sites 7, 8, and 12 support this: Miller et al. (2018) recorded numerous clear ice





Figure 4. (a) Borehole profiles acquired at Sites 7, 8, and 12 (Miller et al., 2018), showing specific discharge (SD), density, stratigraphy, depth to water table, aquifer base (Montgomery et al., 2017), *Vp* (Montgomery et al., 2017), and *Vs* (this study). The flow zone was defined in Miller et al. (2018) by positive specific discharge. (b) Average porosity estimation for all sites using the Biot-Gassmann theory, including water content estimates from Montgomery et al. (2017). Error bars are based on the upper and lower *Vp:Vs* estimates.

lenses at Site 12 (Figure 4a), where the clear ice content increased by 9% and 12% compared to Site 7 and 8, respectively, calculated from 0 to 38 m depth. At Site 12, where the water table depth is only 10 m, Vs is 950 ± 120 m/s in the unsaturated firm and $1,280 \pm 170$ m/s in the underlying saturated layer.

Posterior distributions of the ratio of *Vp*:*Vs* were also output from MuLTI III. The most likely solutions are plotted for all sites (Figure 3b) and average *Vp*:*Vs* values derived for each layer (Figure 3d). Gagnon et al. (1988) show *Vp*:*Vs* ratios of a single ice crystal are 2.14 and 1.97, respectively, for measurements along and orthogonal to the *c*-axis. When observed from seismic data, *Vp*:*Vs* of 2 is typically recorded for standard ice in the upper ice column (Wittlinger & Farra, 2012); hence, this value is used as a reference ratio for ice in this study. Figures 3b and 3d show the average *Vp*:*Vs* of the saturated firn aquifer is 2.30 ± 0.54 in the upslope Sites 1–7 and 2.01 ± 0.46 in Sites 10–12, suggesting that porosity is likely lower in the downstream sites, with the *Vp*:*Vs* ratio closely approximating that of ice.





Figure 5. Schematic interpretation of the firn aquifer system. The enlarged firn images were adapted from Blunier and Schwander (2000).

3.2. Porosity Estimation

Biot (1941) and Gassmann (1951) related the bulk Vp:Vs of a material, $Vp:Vs_{bulk}$, to the velocity ratio of its matrix, $Vp:Vs_{matrix}$, material total porosity (\emptyset), and the pore-filling interstitial fluid (here liquid water). The change in $Vp:Vs_{bulk}$ observed in a porous fluid-filled medium is

$$\frac{1}{\frac{Vp}{Vsbulk}} = \frac{1}{\frac{Vp}{Vsmatrix}} (1 - \emptyset).$$
(2)

As we obtain $Vp:Vs_{bulk}$ from our study and set $Vp:Vs_{matrix} = 2.0$ (Gagnon et al., 1988; Wittlinger & Farra, 2012), equation 2 can be rearranged to yield porosity (Figure 4b). In some cases, our derived $Vp: Vs_{bulk} < Vp:Vs_{matrix}$, resulting in negative porosities, an artifact of $Vp:Vs_{bulk}$ approaching the velocity of solid ice, and thus extending to lower Vp:Vs within uncertainty bounds. Furthermore, equation 2 does not account for differential pressure and compaction (as modified in Lee, 2002) and assumes that pore fluid does not affect the shear modulus of the sediment. Also, equation 2 does not account for velocity anisotropy effects, which may be present through the aquifer. Therefore, further work is required to address these limitations. Here, we report an underestimation for the downslope sites, 10–12, where the likely increase in ice content, particularly at the base (e.g., Site 12; Figure 4a), stiffens the material, hence decreasing Vp:Vs, thus biasing our Biot-Gassmann approach to lower porosities for these sites. Notably, Site 10 implies a negative porosity (-0.04 ± 0.22), indicating poorly constrained Vp:Vs data and potentially having an effective medium property that cannot be described by a Biot-Gassmann approach.

Nonetheless, there is a decrease in aquifer porosity downstream, below 1.65 km elevation, with average porosity 0.11 ± 0.22 in the upslope sites (1–7) and 0.01 ± 0.24 in the downslope sites (10–12), again consistent with the appearance of additional clear ice in cores (Figure 4a). Our porosity estimates generally agree, within error margins, with those of Montgomery et al. (2017), reported as upper-bound estimates of water content, which use the relative change in *Vp* in the aquifer verses the underlying ice.

4. Discussion and Conclusions

This study integrates multiple geophysical observations to evaluate the physical properties of a firn aquifer. Upstream, the infiltration of water to the firn matrix is characterized by a decrease in Vs and progressive increase in Vp:Vs; downstream, the increased clear ice content (9% increase from Site 7 to Site 12) increases Vs and decreases Vp:Vs, bringing it closer to that of pure ice. These additional ice lenses reduce the porosity and storage potential of the firn and potentially facilitate water flow consistent with Miller et al. (2018). From the flow measurements in Figure 4a, the ice lenses do not appear to impede lateral flow within the aquifer but could reduce vertical flow if the ice lenses cover a large area (e.g., Humphrey et al., 2012). These along-flow changes suggest that the aquifer transitions from a flooded firn layer with minor clear ice

lenses upstream to a more heterogeneous aquifer with many clear ice lenses downstream (Figure 5). We interpret the presence of clear ice lenses as being diagnostic of episodic refreezing of the aquifer system, in particular at the aquifer base where water-saturated firn produces clear ice layers (Figure 4a; Miller et al., 2018). Recent studies have suggested that the firn aquifer formed earlier at the lower elevation sites (e.g., Miller et al., 2020), therefore undergoing more cycles of melt and freeze, having a longer history of advection and existence, compared to the upslope sites.

Our observations suggest a decrease in aquifer porosity below 1.65 km elevation: The porosity in the young aquifer may exceed 11% (average from Sites 1 to 7), whereas the mature aquifer may have porosity locally reduced to 1% (average from Sites 10 to 12), notwithstanding our error estimates. Compared to porosities derived from Vp alone (Montgomery et al., 2017), the downslope sites' (10–12) porosities are underestimated in this study, likely owing to the strong dependence Vs has on shear modulus, where the increase of ice in the matrix causes an increase in the shear modulus to that more like pure ice, where the porosity is 0.

This paper presents the first insights into the *Vs* distribution of a firn aquifer. It details how a complete seismic record can be used to extract elastic properties of the aquifer and inform porosity and stiffness. Our novel methodology enables the direct integration of multiple geophysical data sets, providing a comprehensive uncertainty analysis, circumventing ambiguous solutions that are often obtained from individual datasets. We provide an evidenced description of the development of the aquifer highlighting its complex spatial variability and the reduced water storage potential of older systems. This insight has important implications for predictive modeling and understanding existing and new firn aquifers being discovered globally. We recommend a combination of GPR, seismic, and borehole measurements for deriving accurate properties of an aquifer, combining the spatial coverage of geophysical data with the detailed ground-truth of intermittent borehole measurements.

Data Availability Statement

The MuLTI III algorithm can be found at https://zenodo.org/record/4015183 (http://doi.org/10.5281/ zenodo.4015183). The seismic data are archived at the IRIS Data Management Center (GreenISE, Data set 14-064) and the Arctic Data Center (https://arcticdata.io/catalog/view/doi:10.18739/A2RB6W22B).

References

- Bartholomew, I. D., Nienow, P., Mair, D., Hubbard, A., King, M. A., & Sole, A. (2010). Seasonal evolution of subglacial drainage and acceleration in a Greenland outlet glacier. *Nature Geoscience*, 3(6), 408–411. https://doi.org/10.1038/ngeo863
- Biot, M. A. (1941). General theory of three-dimensional consolidation. Journal of Applied Physics, 12(2), 155–164. https://doi.org/10.1063/ 1.1712886
 - Blunier, T., & Schwander, J. (2000). Gas enclosure in ice: Age difference and fractionation. In Physics of Ice Core Records (pp. 307–326). Hokkaido University Sapporo, Japan: Hokkaido University press.
- Chu, W., Schroeder, D. M., & Siegfried, M. R. (2018). Retrieval of englacial firn aquifer thickness from ice-penetrating radar sounding in Southeastern Greenland. *Geophysical Research Letters*, 45, 11–770. https://doi.org/10.1029/2018GL079751
- Diez, A., Eisen, O., Weikusat, I., Eichler, J., Hofstede, C., Bohleber, P., et al. (2014). Influence of ice crystal anisotropy on seismic velocity analysis. Annals of Glaciology, 55, 97–106. https://doi.org/10.3189/2014AoG67A002
- Forster, R. R., Box, J. E., van den Broeke, M. R., Miège, C., Burgess, E. W., van Angelen, J. H., et al. (2014). Extensive liquid meltwater storage in firm within the Greenland ice sheet. *Nature Geoscience*, 7, 95–98. https://doi.org/10.1038/ngeo2043
- Fountain, A. G., & Walder, J. S. (1998). Water flow through temperate glaciers. *Reviews of Geophysics*, 36(3), 299–328. https://doi.org/ 10.1029/97RG03579
- Gagnon, R. E., Kiefte, H., Clouter, M. J., & Whalley, E. (1988). Pressure dependence of the elastic constants of ice Ih to 2.8 kbar by Brillouin spectroscopy. Journal of Chemical Physics, 48(C1), C1-23–C1-28. https://doi.org/10.1051/jphyscol:1987104
- Gassmann, F. (1951). Elastic waves through a packing of spheres. *Geophysics*, *16*(4), 673–685. https://doi.org/10.1190/1.1437718 Gazetas, G. (1982). Vibrational characteristics of soil deposits with variable wave velocity. *International Journal for Numerical and*
- Analytical Methods in Geomechanics, 6(1), 1–20. https://doi.org/10.1002/nag.1610060103 Harper, J., Humphrey, N., Pfeffer, W. T., Brown, J., & Fettweis, X. (2012). Greenland ice-sheet contribution to sea-level rise buffered by
- Harper, J., Humphrey, N., Prener, W. 1., Brown, J., & Fettweis, X. (2012). Greenland ice-sneet contribution to sea-level rise buffered by meltwater storage in firn. *Nature*, 491(7423), 240–243. https://doi.org/10.1038/nature11566
- Hubbard, B., Luckman, A., Ashmore, D. W., Bevan, S., Kulessa, B., Munneke, P. K., et al. (2016). Massive subsurface ice formed by refreezing of ice-shelf melt ponds. *Nature Communications*, 7, 1–6. https://doi.org/10.1038/ncomms11897
 - Humphrey, N. F., Harper, J. T., & Pfeffer, W. T. (2012). Thermal tracking of meltwater retention in Greenland's accumulation area. Journal of Geophysical Research, 117, F01010. https://doi.org/10.1029/2011JF002083
 - Killingbeck, S. F., Livermore, P. W., Booth, A. D., & West, L. J. (2018). Multimodal layered transdimensional inversion of seismic dispersion curves with depth constraints. *Geochemistry, Geophysics, Geosystems*, 19, 4957–4971. https://doi.org/10.1029/2018GC008000
 - Koenig, L. S., Miege, C., Forster, R. R., & Brucker, L. (2014). Initial in situ measurements of perennial meltwater storage in the Greenland firm aquifer. *Geophysical Research Letters*, 41, 81–85. https://doi.org/10.1002/2013GL058083
 - Kuipers Munneke, P., Ligtenberg, S., van den Broeke, M., van Angelen, J., & Forster, R. (2014). Explaining the presence of perennial liquid water bodies in the firm of the Greenland Ice Sheet. *Geophysical Research Letters*, 41, 476–483. https://doi.org/10.1002/2013GL058389

Acknowledgments

This work was supported by Antarctic Science Ltd. and the UK NERC SPHERES DTP (grant no. NE/L002574/1). NS and LM were supported by the National Science Foundation Office of Polar Programs (grant no. NSF1417993). The geophone instruments were provided by the Incorporated Research Institutions for Seismology (IRIS) through the PASSCAL Instrument Center at New Mexico Tech. The facilities of the IRIS Consortium are supported by the National Science Foundation's Seismological Facilities for the Advancement of Geoscience (SAGE) Award under Cooperative Support Agreement OPP-1851037. ArcticDEM provided by the Polar Geospatial Center under NSF-OPP awards 1043681, 1559691, and 1542736. Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government.



- Kuipers Munneke, P., Ligtenberg, S., van den Broeke, M., & Vaughan, D. (2014). Firn air depletion as a precursor of Antarctic ice-shelf collapse. Journal of Glaciology, 60, 205–214. https://doi.org/10.3189/2014JoG13J183
- Lee, M. W. (2002). Biot-Gassmann theory for velocities of gas hydrate-bearing sediments. Geophysics, 67(6), 1711–1719. https://doi.org/ 10.1190/1.1527072
- Legchenko, A., Miège, C., Koenig, L. S., Forster, R. R., Miller, O., Solomon, D. K., et al. (2018a). Estimating water volume stored in the south-eastern Greenland firn aquifer using magnetic-resonance soundings. *Journal of Applied Geophysics*, 150, 11–20. https://doi.org/ 10.1016/j.jappgeo.2018.01.005
- Legchenko, A., Miège, C., Koenig, L. S., Forster, R. R., Miller, O., Solomon, D. K., et al. (2018b). Investigating a firn aquifer near Helheim Glacier (South-Eastern Greenland) with magnetic resonance soundings and ground-penetrating radar. *Near Surface Geophysics*, 16, 411–422. https://doi.org/10.1002/nsg.12001
- Miège, C., Forster, R. R., Brucker, L., Koenig, L. S., Solomon, D. K., Paden, J. D., et al. (2016). Spatial extent and temporal variability of Greenland firn aquifers detected by ground and airborne radars. *Journal of Geophysical Research: Earth Surface*, 121, 2381–2398. https:// doi.org/10.1002/2016JF003869
- Miller, O., Solomon, D. K., Miège, C., Koenig, L., Forster, R., Schmerr, N., et al. (2020). Hydrology of a perennial firm aquifer in Southeast Greenland: An overview driven by field data. *Water Resources Research*, 56, e2019WR026348. https://doi.org/10.1029/2019WR026348 Miller, O., Solomon, D. K., Miège, C., Koenig, L., Forster, R., Schmerr, N., et al. (2018). Direct evidence of meltwater flow within a firm
- aquifer in southeast Greenland. Geophysical Research Letters, 45, 207–215. https://doi.org/10.1002/2017GL075707
- Montgomery, L. N., Schmerr, N., Burdick, S., Forster, R. R., Koenig, L., Legchenko, A., et al. (2017). Investigation of firn aquifer structure in southeastern Greenland using active source seismology. Frontiers in Earth Science, 5, 10. https://doi.org/10.3389/FEART.2017.00010
- Park, C. B., Miller, R. D., & Xia, J. (1999). Multichannel analysis of surface waves. *Geophysics*, 64, 800–808. https://doi.org/10.1190/ 1.1444590
- Parry, V., Nienow, P., Mair, D., Scott, J., Hubbard, B., Steffen, K., & Wingham, D. (2007). Investigations of meltwater refreezing and density variations in the snowpack and firm within the percolation zone of the Greenland ice sheet. *Annals of Glaciology*, 46, 61–68. https://doi. org/10.3189/172756407782871332
- Poinar, K., Dow, C. F., & Andrews, L. C. (2019). Long-Term Support of an Active Subglacial Hydrologic System in Southeast Greenland by Firn Aquifers. *Geophysical Research Letters*, 46, 4772–4781. https://doi.org/10.1029/2019GL082786
- Porter, C., Morin, P., Howat, I., Noh, M. J., Bates, B., Peterman, K., et al. (2018). ArcticDEM, doi.org/10.7910/DVN/OHHUKH, Harvard Dataverse, V1, [accessed on June 2, 2020].
- Reeh, N., Fisher, D. A., Koerner, R. M., & Clausen, H. B. (2005). An empirical firm densification model comprising ice lenses. Annals of Glaciology, 42, 101–106. https://doi.org/10.3189/172756405781812871
- Wathelet, M., Chatelain, J.-L., Cornou, C., Di Giulio, G., Guillier, B., Ohrnberger, M., & Savvaidis, A. (2020). Geopsy: A user-friendly opensource tool set for ambient vibration processing. Seismological Research Letters, 91, 1878–1889. https://doi.org/10.1785/0220190360
- Wittlinger, G. A., & Farra, V. (2012). Observation of low shear wave velocity at the base of the polar ice sheets: evidence for enhanced anisotropy. *Geophysical Journal International*, 190, 391–405. https://doi.org/10.1111/j.1365-246X.2012.05474.x
- Young Kim, K., Lee, J., Ho Hong, M., Kuk Hong, J., Keun Jin, Y., & Shon, H. (2010). Seismic and radar investigations of Fourcade Glacier on King George Island, Antarctica. *Polar Research*, 29(3), 298–310. https://doi.org/10.3402/polar.v29i3.6082